Author's response to reviews

Title:Torsion of the greater omentum. Cases report study

Authors:

Konstantinos Alexiou (knego@otenet.gr)
Argyrios Ioannidis (argioannid@med.uoa.gr)
Ioannis Drikos (johndrikos@yahoo.com)
Nicolaos Sikalias (jdrikos@hotmail.com)
Nicolaos Economou (gramdiy@sismanoglio.gr)

Version:4 Date:17 June 2015

Author's response to reviews:

Editorial requests:

1. Please include the study design in your title, i.e. Case report. For example: A presenting with B in C: a case report.(eg. A case of massive GI bleeding in a patient with 2 small AVMs in the small intestine: a case report.)
Done

2. Please reformat the title page as page 1. The title page should include the title, authors, author affiliations, corresponding author details and complete co-author email addresses.
Done

3. Please enumerate the cases in the Case presentation section accordingly (e.g. Case 1 patient...Case 2 patient...)
Done

4. Please change the description of the patient's gender to 'male/female'.
Done

5. Ethnicity of the patient in the Abstract and Case presentation sections.
Done

6. Please reformat the Authors' Contributions section at the end of the manuscript, before the reference list. We suggest the following kind of format (please use initials to refer to each author's contribution):
Done